Author's response to reviews

Title: Oral and dental health care practices in pregnant women in Australia: a postnatal survey

Authors:

Natalie J Thomas (natalie.thomas@adelaide.edu.au)
Philippa F Middleton (philippa.middleton@adelaide.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)

Version: 7 Date: 17 April 2008

Author's response to reviews: see over
17 April 2008

Dear The BioMed Central Editorial Production Team,


Thank you for your letter of acceptance. We have read through our paper thoroughly and made some minor typographical changes where necessary.

Please do not hesitate to contact us if further changes are needed. We look forward to seeing our paper in print in the near future.

With best wishes
Yours sincerely

Ms Natalie Thomas Professor Caroline Crowther Ms Philippa Middleton
Discipline of Obstetrics and Gynaecology